

# Sign Me Up! A Model and Field Experiment

# on Volunteering

Erte Xiao and Daniel Houser

January 2014

**Discussion Paper** 

Interdisciplinary Center for Economic Science 4400 University Drive, MSN 1B2, Fairfax, VA 22030 Tel: +1-703-993-4719 Fax: +1-703-993-4851 ICES Website: <u>http://ices.gmu.edu</u> ICES RePEc Archive Online at: <u>http://edirc.repec.org/data/icgmuus.html</u>

#### SIGN ME UP! A MODEL AND FIELD EXPERIMENT ON VOLUNTEERING

Erte Xiao Department of Social and Decision Sciences Carnegie Mellon University exiao@andrew.cmu.edu

> Daniel Houser Department of Economics George Mason University dhouser@gmu.edu

#### January 12, 2014

**Abstract:** We develop and model a two-stage incentivized intervention to promote pro-sociality. In the first stage, participants are incentivized to complete a compound task consisting of a targeted pro-social activity and a complement activity. In the second stage, participants are incentivized to complete repeatedly only the complement activity. The model predicts that, conditional on compliance with the first-stage compound task, intrinsic interest in the target activity is promoted regardless of compliance with the second-stage task. To test this we design and implement a field experiment on volunteering. The results are consistent with our model. Moreover, in the one year subsequent to our experiment, those who participated in our compound-task mechanism reported volunteering systematically more than those who participated in alternative mechanisms we investigated. Our approach has useful implications for promoting positive individual and social outcomes in many behavioral domains.

**JEL:** C93; D02; D03; D04; D64; H41; Z13

Keywords: volunteering; incentives; pro-social attitudes; cognitive dissonance; field experiment

Acknowledgements. We thank Karen Meharra and Kaylee Petsch for enabling our cooperation with the Carnegie Library of Pittsburgh. John Cusick, David Hagmann and Shuwen Li provided outstanding research assistance with the data. We thank Jason Aimone, Omar Al-Ubaydli, Fangfang Tan, Viktor Brech and seminar participants at LMU, Munich; University of Cologne; University of Bonn; Max Planck Institute, Jena; Max Planck Institute, Bonn; Peking University, Tshinghua University, Nanyang Technological University, National University of Singapore, University of Wisconsin, Madison; Virginia Tech; the 4th Annual Xiamen University International Workshop on Experimental Economics, the Science of Philanthropy Initiative Conference, the South East Asian meetings of the Econometric Society 2013 and the International Meetings of the ESA 2013 for valuable comments.

#### 1. Introduction

Volunteering refers to "any activity in which time is given freely to benefit another person, group, or organization" (Wilson, 2000). Substantial research has shown that volunteerism not only has substantial social benefits, but also various health benefits (e.g. Piliavin and Siegl, 2007; Okun et al., 2013). Additionally, it enables volunteers to develop skills and knowledge (Wilson and Musick, 1999; Meier and Stutzer, 2008; Borgonovi, 2009). Despite numerous social and personal benefits, participation in volunteering arguably remains low: about 75% of people did not volunteer in 2012 (Current Population Survey, USBLS)<sup>1</sup>. Consequently, substantial effort continues to be directed towards finding ways to promote volunteer activity. In this paper, we model and test using a field experiment a novel compound task incentive-based approach to promoting intrinsic interest in volunteering. The key advantage to our mechanism is that it is "failsafe." In particular, both theoretically and empirically, we show that it can promote interest in the target activity (in our case, volunteering) even when people fail to comply with the rewarded activity. This stands in sharp contrast to conventional incentive-based approaches where non-compliance can lead to reduced desire to engage in the targeted behavior.

Across the social-sciences, scholars have worked towards understanding how to increase the supply of volunteers. The two main approaches involve encouraging people to develop: (i) a favorable intrinsic attitude toward volunteering; or (ii) a habit of volunteering (Janoski et al., 1998; Wilson, 2000). The first approach argues for the importance of building institutions to disseminate positive values and attitudes toward volunteering. The idea is that an intrinsic interest in helping others is a motive for volunteer work by voluntary organizations. Regardless of previous engagement in volunteer activities, those who value helping others are more likely than those who do not to respond positively to volunteer opportunities.

In contrast, social participation emphasizes the importance of habits. Supporters of this approach point to the importance of involving young people in volunteering in order to acclimate them to pro-social routines. Indeed, many organizations, such as United Way<sup>2</sup>, involve high school students in community projects. Likewise, leaders of charity organizations routinely emphasize the importance of giving small amounts when the contributors have little to give (Rosen and Sims, 2011; Meer, 2013). Thus, the habit formation approach encourages the view

<sup>&</sup>lt;sup>1</sup> http://www.bls.gov/news.release/volun.t02.htm

<sup>&</sup>lt;sup>2</sup> http://www.unitedway.org/take-action/youth-volunteering

that volunteering is "what we do," while the former approach encourages the perception that volunteering is "what we want to do."

Both approaches are potentially useful. For reasons described in Section 2 below, our mechanism is designed to promote intrinsic interest in volunteering. Our evidence is that it does so successfully, and that our mechanism promotes volunteering activity.

Our study offers several contributions to the economics literature. First, while mechanisms for promoting giving have been heavily studied, nearly all of this research has focused on giving tangible items, such as money and organs (List and Lucking-Reilley,2002; Roth et al., 2004; Lacetera et al., 2012). To our knowledge, only a few studies have examined mechanisms to promote the giving of time using field experiments, and most of these focus on the effect of volunteer participation during interventions (Gneezy and Rustichini, 2000; Al-Ubaydli and Lee, 2011; Conrads et al., 2013). This study sheds light on the impact of incentive-based interventions on attitudes toward volunteering. Second, we contribute to the literature on the "crowding out" effect of monetary incentives (Deci, 1971; Frey and Jegen, 2001; Fehr and Falk, 2002; Li et al., 2009; also see Gneezy et al., 2011; Bowles and Polania-Reyes, 2012 for reviews) by pointing to another reason rewards can fail: people may become less interested in rewarded activities when they fail to achieve the reward. A key message of our paper is that it is crucial to consider the impact of low compliance rates on intrinsic interest in the targeted activity when designing incentive systems.

Next, we develop an incentive-based approach using compound tasks. The compound task includes both the target task (volunteering in our case) and a complement task (writing about the benefit of volunteering). Building on previous research, we show theoretically that as long as people can complete the compound task once, providing incentives to repeat the complement task can be more effective than the conventional approach where incentives are provided for repeating the targeted activity. The intuition behind our approach is that people learn about their own preferences in the same way that an econometrician learns about agents' preferences from revealed preference data. In typical analyses, an econometrician learns about an agent's preferences in an analogous way, then it should in principle be possible to manipulate the choice set – the alternatives not chosen – and in so doing manipulate preferences. In our case, rewarding the target task is risky in that failure to comply signals disinterest. In contrast, if we

incentivize the complement task, it is less likely that the failure to comply reveals distaste for the target activity. This is the fundamental idea behind our "failsafe" approach. We explain below that, in fact, a failure to complete the complement task will lead people to believe that they prefer the target task.

Finally, we test our proposed mechanism using a field experiment on volunteering. Although our application is volunteering, we expect our framework can be applied successfully to many behavioral domains.

Our results are consistent with the model. In particular, our proposed mechanism is better able to promote intrinsic interest in (preferences for) volunteering than any of the four natural alternative approaches to promoting volunteering we consider. Moreover, during the one year following our intervention, reported volunteering among people who participated in our proposed mechanism was greater than the volunteering reported by those who participated in the four other mechanisms. Consequently, we find that our proposed approach promotes not only intrinsic interest in volunteering but also volunteer activity.

This paper is organized as follows. Section 2 explains why we focus on promoting positive attitudes toward volunteering. We model our compound task approach to attitude change in Section 3. Section 4 describes the experiment designs. We derive the hypothesis of the experiment in Section 5. Section 6 reports our data and results and Section 7 concludes.

#### 2. Intervention target: habit formation (practice) or attitude

Notwithstanding potential crowding-out effects of monetary incentives, several recent studies demonstrate that we can promote targeted activity if people are rewarded for engaging in the activity *repeatedly* and over an extended period (Charness and Gneezy, 2009; Acland and Levy, 2011; Royer et al., 2013). One explanation for this is that repeated engagement creates a habit of practicing the targeted activity. Consequently, it is natural to consider whether monetary rewards might be useful in creating a volunteering habit. In this case, however, there are reasons to worry that incentives may have a detrimental effect. One is that paying people to volunteer may lead people not to view themselves as engaging in volunteering. As a result, even if people comply with an incentivized task over a long period of time, it may not create a habit of volunteering.

Moreover, it may be extremely difficult to volunteer at high frequencies. Whether one is able to volunteer does not merely depend on a desire to volunteer, but on many other exogenous factors, such as schedules, locations of activities, and the amount of time available to the potential volunteer. In particular, many short-term volunteer opportunities can be surprisingly time-intensive, in that they require planning, registering, and sometimes even background checks in advance of the actual volunteer activity. These hurdles reduce the ease with which one can volunteer, and often mean that volunteering is more of a planned than a spontaneous choice.

The complexity of social participation in volunteering contrasts with activities such as exercising, which is easier to implement spontaneously by walking, running or going the gym. Indeed, survey data suggest that differences between the frequency of exercising and volunteering is dramatic. According to Gallup 2012, about 70% of adults exercised more than 30 minutes per week (about 26 hours a year) in 2011. In contrast, survey data from the BLS (2012) indicates that only about 20% of adults volunteered more than 15 hours in 2012. This raises the concern that people may fail to comply with rewarded high-frequency volunteering activities.

It is important to note that in previous studies on reward mechanisms, participants succeeded in completing the rewarded task. For example, Charness and Gneezy (2009) report that over 90% of participants completed the exercise task and received the reward. Thus, their findings cannot inform the effect of reward when compliance rates are low. The worry is that when incentives fail to generate compliance, crowding-out may occur.

Recent studies in psychology and behavioral economics suggest that, in contrast to the neo-classical economic paradigm, people's actions are not always driven by well-structured preferences. Instead, these studies suggest that people infer their preferences (positive or negative) from the results of their previous actions or memory (DeJong, 1979; Benabou and Tirole, 2011; Ariely and Norton, 2007; Weber and Johnson, 2006). In light of these studies, we hypothesize that failure to comply with the rewarded task may lead people to believe that they do not really like to perform the task. This may have a detrimental effect on their willingness to perform the targeted activity.

In addition to the potential detrimental effect of reward, the relatively low-frequency of volunteering also suggests that it may be difficult to measure the immediate behavioral impact of incentive interventions. This raises a practical challenge for researchers hoping to investigate the effectiveness of interventions. Indeed, due to this concern, some private-sector groups working

5

to improve pro-social behavior instead use expressions of interest as their metric to evaluate different programs<sup>3</sup>.

In view of these concerns, we decided to focus on designing mechanisms to promote positive attitudes toward volunteering. Positive attitudes in this regard are surely important. For example, studies have shown that children who develop a positive attitude toward volunteering are more likely to volunteer later in their adult life (Astin 1993; Damico et al 1998; Janoski et al., 1998). Knowing how interventions influence attitudes of course requires a reliable attitude measure. Towards this end, we measure attitude by examining participants' choices of whether to sign up to receive information for future volunteering opportunities. The reason is that one is more likely to sign up when one is more interested in volunteering. Although signing up does not necessarily imply that one will volunteer organizations strive to obtain people's contact information and ask them to sign up to receive volunteering information.

In the next section, we model a two-stage incentivized intervention based on a "compound task" to promote a positive attitude toward volunteering. We elaborate how this approach can be "failsafe" in that it can promote intrinsic positive attitude toward volunteering regardless of whether participants comply with the second-stage rewarded task. As a result, it avoids the potential detrimental effect of reward when compliance rates are low.

#### 3. A Compound Task Approach for Attitude Change

Our proposed compound task approach to promote a positive attitude towards a target activity is based on one of the most influential constructs in social psychology: cognitive dissonance theory (Festinger and Carlsmith, 1959). Cognitive dissonance theory famously illustrates how actions can create rather than reflect preferences (Ariely and Norton, 2007).

In their experiment, Festinger and Carlsmith (1959) asked participants to complete a boring task. Subsequently, the experimenter requested that they convince a new participant (actually a confederate) that the task was interesting. In one condition, participants were paid \$1 to lie to this person, and in another condition \$20. Finally, all participants were asked to rate their enjoyment of the original (boring) task. Festinger and Carlsmith found that those in the \$1 group

<sup>&</sup>lt;sup>3</sup> For example, Jon Behar's website (http://www.apaththatsclear.com/) uses giving games to promote charitable giving, and uses rates at which people sign-up to receive information about giving opportunities as the key measure of his intervention's success.

rated the boring task substantially more positively than those in \$20 group. They proposed cognitive dissonance theory to explain the results.

In brief, cognitive dissonance theory argues that people hold the desire to maintain consistency between their beliefs and their actions. When one is paid a very small amount (in their case \$1) to lie about the boring task, one cannot use that amount to justify the lie. Thus, participants form the belief that they actually did enjoy the task (as compared to forming unpleasant belief that they are cheap liars). In contrast, a larger payment of \$20 is sufficient to justify the lie (at least in the late 1950s). Consequently, participants in this condition did not need to change their beliefs in order to maintain consistency with their actions.

These results were influential. A substantial literature has now emerged showing that one can design paradigms to promote positive attitudes or interest towards any number of activities (Kitayama et al., 2004; Cialdini and Goldstein, 1984). An example is the "Effort Justification Paradigm." The idea is that when one engages in a strenuous activity that s/he would not typically choose, s/he develops the perception that the activity is attractive in order to justify the effort. Axsom and Cooper (1985) successfully applied this paradigm to induce weight loss. Recently, economists have used consistency to explain economic behaviors (Falk and Zimmermann, 2013; Ellingsen and Johannesson, 2004; Matthey and Regner, 2011)

Our compound task mechanism is a novel but straightforward application of cognitive dissonance theory: it is built on the idea that people desire to maintain consistency between their beliefs and actions. We revise previous incentive approaches that promote interest in activities when people comply with incentives. Our goal is to avoid detrimental effects on interest stemming from failures to comply. We show that one can avoid risks stemming from low compliance by rewarding both a target activity and a "complement" task. We elaborate below on how to use this "compound task" as part of a broader approach to incentivizing intrinsic interest in the target activity.

## 3.1 The Model

Previously studied approaches to promoting positive behavioral changes, such as those described by Charness and Gneezy (2009), employ two stages. The first stage offers a short experience with the target activity, and the second stage a longer opportunity for repeated exposure to the target activity. We do the same, with modifications to circumvent potential detrimental effects stemming from a failure to comply with second-stage incentives.

The essence of out compound task approach is that by offering second-stage rewards for repeating the complement task instead of the target task, a failure to comply no longer results in detrimental effects on attitudes towards the target activity. Instead, a failure of compliance generates an increased preference to engage in the target activity.

Like others, our compound task approach consists of two stages. In Stage 1, individuals are provided an incentive to participate in a one-time compound task. The compound task consists of both the target task and a complement task. The target is the task towards which one hopes to promote a positive attitude. The complement task should be chosen to enhance the positive attributes of the target task. In our case, the target task is volunteering. The complement task can be, for example, writing about or reading articles about the benefits of volunteering. A strong complement task might involve informing friends about the benefits of volunteering. In Stage 2, individuals are rewarded if they succeed in repeating the complement task for a certain period of time. To illustrate how this two-stage incentivized intervention can promote positive attitudes toward the target task, consider the following framework.

Let *T* denote the target activity, and *C* the complement task. Assume that *C* has been chosen so that performing it promotes interest in *T*. A person is incentivized to perform a compound task comprised of *T* and *C*; denote this by *TC*. Successfully completing *TC* results in money earnings  $M_{TC}$ . Denote the psychic value of completing the task as U(T,C). Assume preferences are additively separable between T and C, so that:

$$U(T,C) = U_T(T) + U_C(C)$$

To ease notation, below we omit the "T" and "C" subscripts from the utility function.

Next, assume further that the total value V(TC) to completing the compound task successfully is:

$$V(TC) = U(T,C) + M_{TC} = U(T) + U(C) + M_{TC}$$

Without loss of generality, assume the value of the outside option (not completing TC) is zero.

Drawing on the psychology literature mentioned above, an agent is motivated to maintain consistency between her beliefs and actions. If an agent performs TC then she will hold the belief that she prefers TC to the outside option (that is, she knows the sign of V). However, the action of performing TC does not reveal the values of the components that comprise this utility.

Specifically, an agent infers her preferences by observing her own choices, in a way that is analogous to an econometrician inferring features of individual preferences from choice data. An agent knows she would choose to complete the compound task if and only if V(TC)>0, or:

$$U(T) + U(C) + M_{TC} > 0$$
 (1)

Suppose a person completes this task.<sup>4</sup> He/she thus knows that condition (1) holds although he/she does not know how U(T) and U(C) contribute individually to V(TC).

An agent is next given an opportunity to complete only the complement task C. Suppose the agent knows they will receive money earnings  $M_C > M_{TC}$  for successful completion of this task, and denote their psychic value for completing the task as U(C). Again, suppose the outside option has value zero. Assume the agent knows he/she would complete the complement task if and only if

$$U(C) + M_C > 0 \tag{2}$$

Recall the complement task has been designed so that completing it leads to increased positive feelings toward the target activity. Consequently, by construction, successful completion of the complement tasks promotes interest in the target activity.

Consider now the implication of failure to comply with the complement task. An agent who fails to comply with the complement task knows that (2) does not hold. That is:

$$U(C) < -M_C < -M_{TC} \tag{3}$$

Combining (3) and (1), this agent discovers that:

$$U(T) > - U(C) - M_{TC} > 0$$
 (4)

Thus, an agent who fails to complete task C discovers she prefers task T over the outside option. Consequently, regardless of compliance with C, she develops a positive attitude toward the target activity. For this reason, we say the mechanism is "failsafe."

As a practical matter, as is the case in the present paper, the complement task in Stage 2 will be a repetition of the complement task used in Stage 1. Denote by C' the second stage repetition of the initial complement task C. Suppose the utility for performing C' can be expressed by:

$$U(C') = \alpha U(C) \qquad (5)$$

where  $\alpha > 0$ .

<sup>&</sup>lt;sup>4</sup> The model makes no prediction in cases where a person fails to complete the first-stage compound task.

The lower bound of zero imposes the condition that the sign of the utility will not change as a result of repetition. So, for example, without additional incentives such as payment for a complement task, a person who prefers the outside option to the writing task for a short time will not prefer writing for a long time over the outside option.

We now have:

a) From (1), compliance with Stage 1 implies:

$$U(C) > -U(T) - M_{TC}$$

b) Failure to comply with Stage 2 incentives implies

$$U(C) < -M_C/\alpha \tag{6}$$

Combining a) and b) reveals:

$$U(T) > M_C / \alpha - M_{TC} \quad (7)$$

Now, when  $\alpha \leq M_C / M_{TC}$ , then from (7) an individual draws the inference that: U(T) > 0. In particular, a straightforward application of Cognitive Dissonance theory reveals that, in order to maintain consistency between behaviors and beliefs, one draws the inference that the target task *T* is preferred to the outside option.

On the other hand, if  $\alpha > M_C/M_{TC}$  then  $M_C/\alpha - M_{TC} < 0$ . In this case, either U(T) > 0 or U(T) < 0 can be consistent with condition (7). Thus, an individual cannot form inferences about preferences for the Target activity in relation to the outside option, and consequently the two-stage intervention is neutral. In particular, the mechanism implies no impact on one's beliefs regarding T when one fails to comply with second stage incentives. In this sense, the mechanism is "failsafe."

Overall then, so long as the mechanism is designed so that  $\alpha \leq M_C/M_{TC}$  for a nonnegligible fraction of the population of interest, then the compound task approach is predicted to promote a positive attitude toward *T*.

Three points are worth noting. First, for ease of exposition, the derivation above assumes additively separable utility (although the result holds more generally). For example, if utility takes form  $U(T,C)=U_C(C)U_T(T)$ , then it is easy to see that by taking logs, an analysis exactly analogous to the above can be constructed. Second, in some (perhaps most) cases, the researcher may not know the value of " $\alpha$ " above. As a practical matter, in order to have confidence that  $\alpha \leq M_C/M_{TC}$  is satisfied for a non-negligible fraction of the population, the incentive for compliance with the second stage task  $M_C$  should typically exceed that paid for compliance with the first-

stage compound task  $M_{TC}$ . In the experiment reported below, the second-stage incentive is twice the value of the first-stage incentive.<sup>5</sup> Finally, as in a typical structural econometric analysis of choice data, agents preferences are revealed (in this case not to an econometrician but rather to themselves) both from what they choose to do as well as what they do not. Consequently, agents who make the same choice from different choice sets can form different inferences about their own preferences. The key idea underlying our approach is that one can manipulate preferences by manipulating the choice sets.

### 4. Experiment

Our two stage approach extends the two-stage mechanism designed by Charness and Gneezy (2009) to promote exercise. Their mechanism involves an incentivized one-time first-stage activity followed by a second-stage in which the activity is repeated over time. Like their study, we incentivize the target activity in the first stage (in our case, volunteering). Our mechanism pairs the target activity with a "complement" activity in the one-time first-stage activity, and then incentivizes repeated "complement" activities only in the second stage. As we explained above, this insures against the possibility of detrimental effects stemming from non-compliance during the second-stage task. To test our hypothesis that our proposed mechanism performs better than traditional incentive mechanisms that reward only target activities, we design another two treatments where repeated volunteering is rewarded in the second stage. In yet another treatment, subjects are only incentivized to perform the one-time first-stage activity. The reason for this treatment is that it allows us to draw inferences regarding the effect of the second stage intervention. Finally, we conducted a baseline treatment in which information regarding the value of volunteering was provided but no incentives were offered.

#### 4.1 Target activity: Volunteering

Our goal is to promote intrinsic interest in volunteering. To ease participation in our study, we chose a task that required little advance planning. We provided our participants the opportunity to volunteer at the Carnegie Library of Pittsburgh (CLP) as part of one of three programs:

<sup>&</sup>lt;sup>5</sup> We used a second-stage incentive valued at twice the first-stage because it seemed to be a large difference and was also budget feasible in the event of high rates of second-stage compliance.

"Adopt a shelf" (at the local library); "Shipping and Delivery" (at the CLP shipping center); or "Street Marketing" with any local business (Appendix B gives details of each choice).

Participants needed no special skills to volunteer at CLP, and all three activities were available on a flexible schedule. Participants could choose among multiple locations, though the CLP main library is located on the University of Pittsburgh campus, and this turned out to be the most popular location choice. Another reason we chose the CLP is their capacity to enroll volunteers: during the period of our study the CLP had hundreds of volunteer opportunities available. Finally, an important advantage of the CLP is that they do not require long-term commitments. That is, volunteering even once does much good for the CLP, and there is no detrimental impact on the CLP when volunteers do not return.

All CLP volunteers are required to sign in when they begin and sign out before they leave. These sign-in sheets serve as our record of volunteering activities. To ensure anonymity among the volunteers, each participant created a user name on the recruitment day in the lab and signed in at volunteering sites using only their unique user name.

#### 4.2 Complement task

The complement task required only that the participant email the experimenter a line about the benefit of volunteering. We chose this task because we expected that engaging in it repeatedly would reinforce interest in the target activity, and in particular, potentially make salient both the possibility of volunteering, and its social and private benefits<sup>6</sup>.

The writing task was not intended to be difficult or time consuming. Participants were to simply email their message to a particular email account set up for this experiment. The benefits they listed could be written in their own words or copied (cut and pasted) directly from the web. Participants could even copy the benefits from the information sheet we distributed to all participants detailing social and individual benefits of volunteer activity. They were given no length requirement.

We viewed writing as a natural complement task, though other tasks might also be effective. For example, instead of writing to the experimenters, participants might alternatively

<sup>&</sup>lt;sup>6</sup> Psychologists have long studied the effect of writing on mental health and behavior (*e.g.*, Pennebaker, 1997). Writing is also widely used in psychology to make certain aspects of an environment more salient or accessible to subjects. For example, Weber et al. (2007) show that subjects who were asked first to provide arguments favoring immediate consumption displayed greater discounting than those who had first to provide arguments favoring delayed consumption.

be asked to send messages to a friend, or perhaps spend time reading articles describing the benefits of volunteering. It would be useful to investigate, both theoretically and empirically, what types of complement tasks might be more or less effective in promoting one's interest in volunteering.

#### 4.3 Treatments

Our experiment consisted of five treatments. In *Baseline*, participants were not rewarded for any activities. Participants in *Baseline* were provided information regarding volunteering opportunities at CLP and were invited to participate. To maintain symmetry with the other four treatments, participants were also invited to write to the experimenter their thoughts regarding the benefits of volunteering<sup>7</sup> (see Appendix A for instructions).

In each of the remaining four treatments, all participants were rewarded \$25 for completing two specific tasks (what we call a compound task) within one week. The compound task included: (i) volunteering at least one hour at CLP (the target task); and (ii) writing two messages about the benefit of volunteering (the complement task; see Appendix A for instructions). The four treatments varied after the first week. In *VW1* (volunteering and writing just one time), participants were encouraged to continue to help CLP, but were not rewarded for doing so. In the remaining three treatments, participants were rewarded \$50 (in addition to the original \$25) to complete various tasks.

In the W treatment (described below), participants were rewarded for completing the complement task multiple times. Our model predicts that this is a "failsafe" mechanism in that it can create a positive attitude towards volunteering regardless of compliance. In the other two treatments (also described below), rewards were provided for volunteering or volunteering and writing. As noted above, rewarding the target activity may be ineffective when compliance rates are low.

## W treatment (Failsafe mechanism)

In this treatment, following the \$25 week, participants were rewarded by \$50 to email three messages (any length) about any benefit of volunteering in each of the following four weeks. The only requirement was that the three messages needed to be sent on different days and include

<sup>&</sup>lt;sup>7</sup> It turns out no one wrote any message in the baseline.

different content. This incentivized participants to consider the benefit of volunteering repeatedly over four weeks. It is worth emphasizing that, as noted above, this task was not difficult. The purpose of the task was not to *discover* benefits of volunteering, but rather to motivate *thinking* about volunteering in a positive way. This feature of the writing helped minimize the possibility that people would believe they had failed due to the difficulty in coming up with benefits of volunteering, in turn minimizing the risk that they would draw negative inferences about volunteering.

#### V treatment

Following the \$25 week, participants in this treatment had the opportunity to earn \$50 if they volunteered four times in four weeks. This treatment is closely analogous to Charness and Gneezy (2009)'s exercise task. Participants could volunteer any time during the four weeks, with the requirement that each time they needed to volunteer for at least one hour.

## VW treatment

Participants in this treatment had the opportunity to earn \$50 if they successfully completed the compound task four times in four weeks. That is, to receive \$50, participants needed to volunteer four times (in total) and write two messages per week over four weeks.

#### 4.4 Procedures

We recruited all the 264 participants on the same day to ensure that all participants had identical volunteer opportunities available. During the first day, participants were asked to complete a wide-ranging questionnaire and were given an information sheet describing the benefits of volunteering. Volunteer opportunities at Carnegie Library of Pittsburgh (CLP) were then provided to them. We then encouraged them to volunteer at CLP, and offered them \$2 for completing an application prior to their departure from the lab. In the application form, subjects were also asked to create their own user ID for signing in whenever they volunteered. Among the 264 subjects, 31 were randomly assigned to *Baseline*. The other 233 were offered \$25 if they completed the compound task.

Subjects who completed the \$25 task were asked to return to the lab seven days later to receive their payment. We then randomly assigned the remaining 233 subjects to the four reward

treatments (VW1, W, V and VW). The randomization was constrained to ensure that each treatment included similar numbers of participants who successfully completed the \$25 task. This procedure resulted in 58 subjects assigned to the VW1, V and W treatments and 59 assigned to the VW treatment. Each treatment included 26 to 28 people who had successfully completed the \$25 task (see Table 1).

Participants learned of the second phase of the experiment only on the seventh day, and not before. In particular, when picking up their \$25, those who were assigned to W,V or VW were told how they could now earn a \$50 reward. Subjects who were randomly assigned to the VW1 treatment received messages that encouraged them to continue to volunteer at CLP. Participants who failed to comply with initial week's \$25 task, and thus did not return to the lab to receive payment, received their treatment assignment via email. The email used language identical to that used for participants who succeeded (see Appendix C for the messages given to participants in each treatment).

Four weeks later, subjects who completed the tasks returned to the lab and received their \$50 earnings. Then, about three weeks later, we sent an email to all participants with a link to a survey. At the conclusion of this survey, participants were asked whether they would like to sign-up to receive information about volunteer opportunities at CLP. There was no default answer: they could click "yes" or "no." Their answer to this question served as a revealed preference measure of their attitude toward volunteering.

The reason we chose to send the survey several weeks after the conclusion of the experiment is that while immediate attitude changes are interesting, of greater practical importance is whether any attitude changes persist. Three weeks is long enough to demonstrate persistence, while still sufficiently temporally close to the intervention to have confidence that any observed changes are due to the experiment. Another reason not to delay beyond three weeks is that response rates may fall as time between the experiment's end and the survey administration increases. In our case response rates were strong, with answers provided by 160 subjects (61% response rate).

#### 5. Hypothesis

Applying the model discussed in section 3 to *W*, consider those who complete the \$25 task in the first week (U(T)+U(C)+25>0). If they complete the writing task in the following four weeks and

thus earn the \$50 reward, repeatedly thinking and writing about the benefit of volunteering will leave the benefits of volunteering more accessible, and thus leave them more likely to hold a positive attitude towards volunteering (Schwarz,1998). On the other hand, if they fail to write for \$50, then they will believe that they do not prefer the writing task to the outside option  $(U(C)+50/\alpha < 0)$ . To reconcile their completion of both volunteering and writing in the first stage with their failure to write in the second stage, they will hold the belief that they prefer volunteering to the outside option as long as  $\alpha \le 2$ .<sup>8</sup> Thus, the reward mechanism in *W* is failsafe in the sense that it can lead individuals to develop a positive attitude towards volunteering regardless of their second-stage compliance rate.

In *V* and *VW*, volunteering is included in the \$50 reward task. Consider those who complete the first-stage \$25 task during the first week. If subjects succeed in the \$50 task, and participate in volunteering repeatedly for four weeks, the repeated participation could lead subjects to become interested in volunteering (e.g., Charness and Gneezy, 2009). On the other hand, if subjects fail to complete the \$50 task, in the *V* treatment they may conclude that they do not prefer volunteering to the outside option. In *VW*, subjects who do not earn the \$50 reward may believe they prefer neither volunteering nor writing to the outside option. Thus, the reward mechanism is risky in both *V* and *VW*. Participants may develop a positive attitude toward volunteering if they comply with the task, but develop a negative attitude if they fail to comply.

Because volunteering is a low frequency activity and we anticipate low compliance rates, we hypothesize that the reward mechanism in W is more likely to promote a positive attitude toward volunteering than either V or VW. That is, we predict that more participants will sign up to receive volunteering information in W than in either V or VW.

### 6. Results

Participants could choose among three volunteering opportunities. Nearly all participants chose "Adopt a shelf" on the University of Pittsburgh library campus: among the 108 participants who completed the compound task in the first stage, two chose "Shipping and Delivery" and seven chose "Street Marketing." In the second-stage, all but one participant who volunteered at all choose "Adopt a shelf." Table 1 details compliance rates for each stage of each condition.

<sup>&</sup>lt;sup>8</sup> As we point out in the Model section, those with a>2 are not predicted to hold any particular belief regarding their preference for *T* in relation to the outside option as a result of a failure to comply with writing during the second stage task.

In Baseline, where no reward was provided, only one participant volunteered. As shown in Table 1, in the \$25 reward conditions, 107 out of 233 subjects volunteered for at least for one hour. Compliance rates were much lower when repeated activities were required to earn \$50: we found 24% in V; 10% in VW and 7% in W. 9 Note these rates differ greatly from those reported in Charness & Gneezy (2009), where compliance rates exceeded 90% and habit formation occurred. Further, since compliance rates are quite similar in VW and W (10% vs. 7%, Z-test, two tail p>0.5) it is interesting to compare sign-up rates between these two conditions. Finally, note the compliance rate in *V* is significantly higher than either *VW* or *W* (Z-test, two tail p<0.05). To test our hypothesis, below we focus on the sign up rate across different treatments.

## 6.1 Email Sign-Up Rate

We first report treatment differences discovered with non-parametric tests and then proceed to discuss results based on regression analyses. In short, we find systematic evidence that W is indeed a failsafe way to promote interest in volunteering.

### 6.1.1 Non-parametric analysis

Our goal is to discover the impact of incentive mechanisms on intrinsic interest in volunteering, and, in particular, whether our proposed failsafe mechanism – illustrated here with the Wtreatment – can promote positive attitudes toward volunteering even under (very) low compliance rates. One hundred sixty subjects completed the survey that ended with the sign-up opportunity. Figure 1 plots the overall signup rate among those who completed the survey in each treatment. The numbers in parentheses (on the x-axis) are failure rates. Consistent with our hypotheses, W achieves the greatest sign-up rate and is the only incentive treatment that achieves significantly greater sign-up rates than the baseline treatment (53% vs. 24%, Z-test, one tail pvalue =0.02). Moreover, the signup rate in W is significantly higher than VW (53% vs. 21%, Ztest, one tail p-value <0.01), which is especially interesting in view of the similarly low compliance rate found in each. The sign-up rate in W is also substantially higher than in V and VWI, although with our sample sizes these differences are not statistically significant at

<sup>&</sup>lt;sup>9</sup> The extremely low compliance rate in W treatment may be somewhat surprising. It would be interesting to understand why most of our participants fail in this seemly easy task. Our messages suggest that most of our participants did not try to succeed. Only four participants successfully completed the writing task; and only three others sent any messages at all. One explanation for this lack of interest is that people may find the task not meaningful (Ariely et al., 2008) despite our framing the task as "share with us your own thoughts on volunteering".

conventional levels (53% vs. 39%, Z-test, one tail p-value=0.14; 53% vs. 38%, Z-test, one tail p-value=0.11).

The failsafe hypothesis is conditioned on participants complying with the first-stage onetime compound task. We next examine sign-up decisions among this particular set of participants. (Recall that each participant who completed the first-stage compound task was randomly assigned to one of the four incentive treatments at the end of the \$25 week.) Among this group, the difference between W and all other incentive treatments is much more apparent. Moreover, these differences are not present among those who failed to comply with the first-stage compound task.

Figure 2 reports separately the sign-up rates of those who succeeded or failed to comply with the compound task. Panel a) reveals that sign-up rates in *W* are significantly higher than *V*, *VW1* and *VW* (68% vs. 42%, Z-test, one tail p-value=0.05; 68% vs. 43%, Z-test, one tail p-value=0.05; 68% vs. 43%, Z-test, one tail p-value<0.01). In contrast, panel b) shows that for those who did not comply with the \$25 task, there are no significant differences between *W* and any other condition (Z-test, one tail p-value>0.10 in all cases).

Next, among those who completed the first-stage compound task, we compare across treatments the sign-up rates for those who fail to comply with \$50 task in the second stage. Sign-up rates are described by Figure 3. Panel a) shows that those participants who failed to comply during the second stage of *V* and *VW* are also significantly less likely to sign up than those who failed to comply in W(25% vs. 69%, Z-test, one tail p-value=0.02; 21% vs. 43%, Z-test, one tail p-value=0.06]. In contrast, those who failed to comply in *W* are marginally significantly more likely to sign-up than are those in *VW1* (69\% \text{ vs. } 43\%, \text{ one tail p-value=}0.06). These comparisons differ sharply from those reported panel b), which reports sign-up rates among those who complied with both the first-stage and second-stage tasks. Sign-up rates among treatments are not significantly different and, in view of our randomization procedures, there is every reason to expect this would be the case.

Figure 3 explains why, among those who succeeded in the compound task in the first stage, the sign-up rate *V* (Figure 2, panel a) is not as low as found in *VW*: compliance in the

second stage is higher in *V* than *VW*. In particular, 11 out of 19 subjects in V who completed the \$25 task and answered the survey also succeeded in the \$50 task. This sharply contrasts with *VW*, where only five of 24 who succeeded in the \$50 task.

A common question is whether selection effects hinder the interpretation of treatment differences revealed in Figure 3. In particular, several have suggested that sign-up rates vary because, for example, those who fail to comply in W are simply more likely to be interested in volunteering than those who fail in VW. It turns out, however, that simple selection effects cannot easily explain the patterns in Figure 3.

To see this, suppose a fraction p ( $0 \le p \le 1$ ) of participants hold a pre-existing willingness to volunteer, and a fraction q ( $0 \le q \le 1$ ) hold a pre-existing willingness to write about volunteering. Suppose further that these characteristics are independently distributed across our participants, and finally that people sign-up for informational emails only if they have a pre-existing interest in volunteering (that is, assume our intervention has no impact on intrinsic interest). Then, it is easy to show that the sign-up rate among those who failed in the second stage should be ordered according to V < VW < W=VW1, where the sign-up rates are predicted to be 0 for V, p(1-q)/(1-pq) for VW, and p for both W and VW1. This is inconsistent with the data reported in Figure 3(a). Further, straightforward selection predicts that the sign-up rate among those who succeed in the second-stage task should be ordered according to VW1 = W < VW = V, with rates of p for VW1 and W, and rates of 100% for V and VW. While the numbers of observations are few, Figure 3(b) does not seem consistent with this ordering. Consequently, simple selection does not seem to offer a plausible alternative explanation for the patterns described in Figure 3<sup>10</sup>.

Note finally that some have suggested that those who complete the first stage compound task may be relatively more sensitive to cognitive dissonance effects than those who do not, and thus more likely to respond to our second-stage manipulation. This may be the case, and if so seems to be a distinct benefit to our approach. In particular, such selection would have the practical advantage of enabling practitioners quickly to focus attention on groups whose attitudes are more likely to be open to change.

#### 6.1.2. Regression analysis

<sup>&</sup>lt;sup>10</sup> It may be possible to produce a more complex selection effect consistent with our data, but this would seem not to have any explanatory advantage over our straightforward model based on consistency.

We conducted a regression analysis to compare individual sign-up decisions across treatments. The independent variables include treatment dummies and gender. The regression results are reported in Table 2. Including all participants who answered the survey (regression (1)), the coefficient of the treatment dummy *W* is significantly higher than that of *VW* and *Baseline* (chi-square test, one-tail, p=0.01 and 0.04, respectively) and also marginally significantly higher than that of VW1 (chi-square test, one-tail. p=0.10). Although the coefficient of *W* is higher than that of *V*, the difference is not significant ((chi-square test, one tail p=0.12). When we include only those who completed the first-stage compound task, the coefficient of *W* is again significantly higher than that of *VW* and *Baseline* (chi-square test, one-tail p<0.01) and marginally significantly higher than that of *VW* and *Baseline* (chi-square test, one-tail p<0.01) and marginally significantly higher than that of *VW* and *Baseline* (chi-square test, one-tail p<0.01) and marginally significantly higher than that of *VW* and *Baseline* (chi-square test, one-tail p<0.01) and marginally significantly higher than that of *V* and *VWI* treatments (chi-square test, one-tail p=0.08 and 0.06, respectively). Overall then, the regression analysis is consistent with the non-parametric results reported above.

Previous research finds females to be more active in volunteering than males (Menchik and Weisbrod, 1987). Consistent with this, our regression results also suggest females are more likely than males to sign up. In the next section we further explore gender differences in our experiment.

#### 6.2 Gender differences

Figure 4 reports sign-up rates by gender in each of our treatments. Overall, female participants are substantially more interested in receiving information about volunteering than are males. Treatment effects are most pronounced for males. For example, comparing *VW* and *W*, both of which have low second-stage compliance rates, male participants are significantly less likely to sign-up in *VW* than in *W* (33% vs. 8%, Z-test, one tail p =0.03). Moreover, the sign-up rate for males is significantly higher in *W* than *VW1* (33% vs. 7%, Z-test, one tail p =0.04).

Recall our failsafe hypothesis is conditioned on participants complying with the \$25 task. We find significantly more female complete the \$25 task then male (53% vs. 29%, Z-test, one tail p<0.01). As a result we have a relatively small number of males who finished \$25 task and filled out the survey with the signup question: 7 subjects in VW1 and V treatment; 4 in W treatment and 11 in VW treatment. Among these few observations on male participants, our main results still hold: male participants who completed the \$25 task have the highest sign up rate in W and lowest in VW (75% vs. 10%, Z-test, one tail p<0.01). On the other hand, for the

female participants who completed the \$25 task, the signup pattern across treatments is very similar to the overall pattern reported in Figure 2(a). In particular, female participants who completed the \$25 task are more likely to sign up in W than VW (67% vs. 38%, Z-test, one-tail p=0.07)

#### 6.3. Volunteer Activity

During late September of 2013 we also conducted another survey to obtain evidence on participants' actual volunteering activity during the year subsequent to our intervention. Participants were asked to report the total number of hours they volunteered in all ways during the period since the study ended. Participants earned \$2 for completing the survey. Below we report data from 45 participants who both responded to our survey and who completed the \$25 task, as well as 10 respondents who participated in our Baseline condition<sup>11</sup>.

Figure 5 details the self-reported volunteering hours of those who completed \$25 task in the four incentive treatments.<sup>12</sup> Figure 5 shows that the pattern of signup rates significantly predicts the pattern of actual volunteering hours (Fisher exact test, p=0.04). In particular, W demonstrates the greatest volunteering hours (21.27) and VW the lowest (10.18). Finally, note that respondents who participated in the Baseline condition reported an average of 12.8 volunteer hours over the course of the year. Taken together, these survey data and signup rates are evidence that W is indeed a failsafe mechanism that can be used effectively to promote even low frequency pro-social activities such as volunteering.

#### 7. Discussion

We report data supporting our hypothesis that people are more likely to develop a positive attitude toward volunteering in W than other treatments. Our metric for intrinsic interest is rates at which people sign-up to receive additional information about volunteering. Consistent with

<sup>&</sup>lt;sup>11</sup> The data reported by those who did not complete the \$25 task are not analyzed for three reasons. First, cell sizes are small, with nine or fewer observations in three of the four treatments. Second, we have no hypotheses regarding these people's behavior. Third, due to self-selection it is difficult to draw conclusions from any differences between groups that did and did not complete the \$25 task.

<sup>&</sup>lt;sup>12</sup> Our analysis excludes reports of greater than 100 volunteering hours. There are two or three such cases in each treatment. The maximum report was 350 (in the W treatment). The reason for this exclusion is that extremely heavy volunteers are more likely to be responding to external incentives unrelated to our interventions (e.g., as indicated by one of our participants, court orders). Including such participants can mask the impact of our mechanism on prosocial decisions.

our hypothesis, we find highest sign-up rates in W, despite the low rates of compliance with W treatment's incentives. This outcome is predicted by a model founded in well-established behavioral economic theory, and whose intuition is simply that people are acting like econometricians for their own preferences. In particular, just as an econometrician draws inferences about agents' preferences by observing which available alternatives are chosen and not chosen, so too does a person uncover their own preferences. Building on this intuition, we show that by manipulating choice sets – alternatives not chosen – one can systematically manipulate people's preferences. Various alternative explanations for our results have been proposed.

One possibility is that subjects in W perceive writing as an easy way to earn \$50. If so, they might sign-up in the hopes of receiving other easy and lucrative future tasks. This explanation is called into doubt for two reasons. First, very few people took advantage of this opportunity to earn money, which suggests people did not perceive it as especially easy. Second, if signups are correlated with the profitability of the task, we would expect the sign up rate pattern across treatments to reflect the pattern of compliance rates. Our data make clear that this is not the case.

A related explanation is that perhaps negative emotions (e.g., guilt, shame or disappointment) were tied to failing a writing task perceived as easy. If so, then people may choose to sign-up as a way to atone for their previous decisions. There are two reasons to doubt this explanation. First, if people sign up to mitigate feelings of guilt or shame, one would again expect relatively high sign-up rates in VW (where guilt can stem not only from a failure to write but also a failure to provide socially beneficial volunteering). As noted, VW received exceptionally low sign-up rates. Second, negative emotions can have many different effects. For example, negative emotions may leave one less likely to take pro-social actions such as signing up to receive information about volunteering. Arguing for a particular impact of a particular negative emotion in a particular treatment seems rather post-hoc, especially in comparison to the theory laid out in the paper.

In sum, we designed a field experiment to test ex-ante predictions made by our behavioral economic model. Our data are consistent with our model's predictions, and we have not yet heard, nor have we been able to develop, an alternative, equally simple, theoretically sound and intuitively plausible explanation for our data.

22

#### 8. Conclusion

Volunteering plays an important role in any human society. Nonetheless, for most of us, volunteering is not an activity that we can easily incorporate into our daily or even weekly routine. This raises a challenge for designing mechanisms to produce increases in volunteering frequency through habit formation. In this paper, we suggest approaching this task from an alternative direction. In particular, our goal was to design a mechanism to promote *intrinsic interest* in volunteering.

Drawing from previous theory in psychology and economics, we first modeled, and then proposed and designed a field experiment to test a "failsafe" mechanism for promoting intrinsic interest in volunteering. Like previous approaches, our mechanism involves two stages and involves incentivizing performance of a targeted activity. Our approach departs from the conventional procedure in that the first stage incentivizes performance of both the target behavior as well as a complement task. The second stage incentivizes behavior of the complement task only. We showed that, conditional on complying with the first-stage activity, this mechanism promotes positive attitudes towards the target activity regardless of second stage compliance.

Our tests yielded empirical evidence that this approach indeed holds promise. In particular, we found that people who complied with the first-stage activities demonstrated an increased interest in volunteering regardless of their compliance with the second-stage task. The increase in positive attitude was greater than that found in participants who were not given the opportunity to perform a second-stage task. Further, the increase was greater than found for those who were given a second-stage incentive to perform the target activity. The reason is that failure to comply with the target activity in the second-stage has a detrimental impact on attitudes toward that activity.

Our study used the sign-up rate as a metric of people's intrinsic interest in volunteering. In practice, volunteer organizations often make substantial efforts in recruiting sign-ups for newsletters or other forms of announcements that help to ensure people have the information they need to take advantage of opportunities for pro-social activities. Our findings may provide insights useful for volunteer organizations interested in designing new ways to promote interest in their activities. Doing so could be particularly valuable, in that our survey data indicate the intervention that promotes greater intrinsic interest in volunteering is also more likely to increase participation in volunteering over the course of the subsequent year. In future research, we intend to extend our findings by developing multi-year panels for the purpose of investigating connections among intrinsic interest, volunteering, life-satisfaction and social well-being.

	# of subjects			Register for —first week	Number volunteered	Number complied
	Total	Female	Male	volunteer	(earn \$25)	(earn \$50)
Baseline	31	14	17	9	1	
VW1	58	33	25	41	26	
\$25+\$50						
V	58	34	24	45	27	14
VW	59	21	38	43	28	6
W	58	25	33	43	26	4

 Table 1. Descriptive data of volunteering in each treatment

 Table 2. Probit regression analysis of sign-up rate of those who answered the survey

		Dependent variable: Signup=1 if yes; =0 if no					
	(1) Include all		(2) Include completed	e only those who \$25 task			
	Coef. (s.e)	p-value	Coef. (s.e)	p-value			
Baseline	-0.26 (0.38)	0.49					
VW1	0.09 (0.23)	0.70	0.03 (0.28)	0.91			
V	0.10 (0.26)	0.69	0.05 (0.31)	0.88			
VW	-0.22 (0.25)	0.40	-0.40 (0.31)	0.19			
W	0.49 (0.24)	0.04	0.63 (0.31)	0.04			
male	-1.20 (0.24)	0.00	-0.71 (0.32)	0.03			
# of obs.	160		85				

# Figure 1. Sign-up rate by treatment



Note: The number on each bar is the number of observations in each condition. The percentage in the parenthesis is the failure rate in the \$50 task.





a) Sign-up rate of the participants who succeeded in the \$25 week

b) Sign-up rate of the participants who failed in the \$25 week



Note: The number on each bar is the number of observations in each condition





a) Sign-up rate of the participants who failed in the second stage task

b) Sign-up rate of the participants who succeeded in the second stage task



Note: The number on each bar is the number of observations in each condition\* In VW1 treatment, all participants who succeeded in \$25 week task are included in both panels.



Figure 4. Sign-Up Rate by Treatment and Gender

Note: The number on each bar is the number of observations in each condition. In Baseline, none of the eight male participants who answered the survey signed up for the email.



Figure 5. Self-reported volunteering hours

Note: The number on each bar is the number of observations of volunteering hours in each condition.

#### References

Acland, Dan and Levy, Matthew. 2011. Habit Formation, Naiveté, and Projection Bias in Gym Attendance. working paper.

Al-Ubaydli, O., Lee, M., 2011. Can tailored communications motivate environ-mental volunteers? A natural field experiment. *American Economic Review*, Vol. 101, No. 3, 2011 101(3), 323–328

Ariely, D., Kamenica, E. and Prelec, D. (2008) "Man's Search for Meaning: The Case of Legos" *Journal of Economic Behavior and Organization*. 67(3), 671-677

Ariely, Dan and Norton, Michael (2007), "How Actions Create—Not Just Reveal—Preferences." TRENDS in Cognitive Sciences. 12 (1), 13 – 16.

Astin A. 1993. What Matters in College? San Francisco: Jossey-Bass

Axsom, D. and Cooper, J. (1985) "Cognitive Dissonance and Psychotherapy: The Role of Effort Justification in Inducing Weight Loss" *Journal of Experimental Social Psychology* 21, pp 149-160

Bénabou, Roland J. M. and Tirole, Jean. (2011). "Identity, Morals and Taboos: Beliefs as Assets," *The Quarterly Journal of Economics*, 126, 805-855.

Borgonovi, F. (2009). Doing well by doing good: The relationship between formal volunteering and self-reported happiness. *Social Science and Medicine*, 66, 2312–2334

Bowles, Samuel and Polania-Reyes, Sandra (2012). Economic Incentives and Social Preferences: Substitutes or Complements? *Journal of Economic Literature* 50(2): 368-425.

Charness, Gary, and Gneezy, Uri 2009. "Incentives to Exercise." Econometrica, 77(3): 909-31.

Cialdini, Robert B. and Noah J. Goldstein (2004), "Social Influence: Compliance and Conformity," *Annual Review of Psychology*, 55, 591–622.

Conrads, J., Irlenbusch, B., Reggiani, T., Rilke, R.M. & Sliwka, D. (2013) "How to Hire Helpers? Evidence From a Field Experiment", CGS working paper

Damico A, Damico S, Conway M. 1998. The democratic education of women: high school and beyond. *Women & Politics* 19:1-31

Deci, E. 1971: Effects of Externally Mediated Rewards on Intrinsic Motivation, *Journal of Personality and Social Psychology*, 18, 105–115

DeJong, William (1979) "An examination of self-perception mediation of the foot-in-the-door effect. Journal of Personality and Social Psychology, 37(12), 1979, 2221-2239.

Ellingsen, Tore and Johannesson, Magnus. (2004). "Promises, Threats, and Fairness". *Economic Journal*. 114, 397-420.

Falk, Armin and Zimmermann, Florian (2013) "A Taste for Consistency and Survey Response Behavior". *CESifo Economic Studies* 59(1):181–193

Frey, B. S., and Jegen, R. (2001): Motivation Crowding Theory, *Journal of Economic Surveys*, 15, 589–611

Festinger, L., and Carlsmith, J. M. (1959). Cognitive consequences of forced compliance. *Journal of Abnormal and Social Psychology*,58, 203-211.

Fehr, Ernst and Falk, Armin (2002) "Psychological Foundations of Incentives", *European Economic Review* 46, 687-724

Gneezy, U., Rustichini, A., 2000. Pay enough or don't pay at all. *The Quarterly Journal of Economics* 115(3), 791–810.

Gneezy, U. S. Meier and P. Rey-Biel (2011) "When and Why Incentives (Don't) Work to Modify Behavior." *Journal of Economic Perspectives*, 25, 4, 191-210.

Janoski, T., Musick, M. and Wilson, J., (1998) Being Volunteered? The Impact of Social Participation and Pro-Social Attitudes on Volunteering. *Sociological Forum*. 12(3). pp. 495-519.

Lacetera, Nicola, Mario Macis, and Robert Slonim. (2012). "Will There Be Blood? Incentives and Displacement Effects in Pro-social Behavior." *American Economic Journal: Economic Policy*, 4(1): 186-223.

Li, J., Xiao, E., Houser, D. and Montague, R. (2009) "Neural responses to sanction threats in two-party economic exchange,", *Proceedings of the National Academy of Sciences of the United States of America*. September 29, vol. 106 no. 39, 16835-16840

List, John and Lucking-Reilley, David. (2002). "The Effects of Seed Money and Refunds on Charitable Giving: Experimental Evidence from a University Capital Campaign," *Journal of Political Economy* 110:1, pp. 215–33.

Kitayama, S., Snibbe, A., Markus, H. and Suzuki, T. (2004). "Is There Any "Free" Choice?: Self and Dissonance in Two Cultures". *Psychological Science*. 15, 527-532.

Matthey, A. and T. Regner (2011). "Do I really want to know? A cognitive dissonance-based explanation of other-regarding behavior", *Games*, 2 (2011), 114-135.

Meer, J. (2013) "The Habit of Giving." *Economic Inquiry*. 51(4), pp. 2002–2017.

Meier, Stephan, Stutzer, Alois, (2008)."Is volunteering rewarding in itself?" *Economica*. 75 (297), pp 39–59.

Menchik, Paul L and Weisbrod, Burton A (1987). "Volunteer labor supply" *Journal of Public Economics* 32(2). pp 159–183.

Okun, Morris A.; Yeung, Ellen WanHeung and Brown, Stephanie (2013). "Volunteering by older adults and risk of mortality: A meta-analysis". *Psychology and Aging*, 28(2), 564-577.

Piliavin, Jane Allyn and Siegl, Erica (2007) "Health Benefits of Volunteering in the Wisconsin Longitudinal Study, *Journal of Health and Social Behavior*, 48 (4), 450-464

Pennebaker, J.W. (1997). "Health effects of the expression ( and non-expression ) of emotions through writing", in A.J.J.M. Vingerhoets, F.J.Van Bussel, and A.J.W. Boelhouwer (Ed.), The (non)expression of emotions in health and disease, Tilburg, Netherlands: Tilburg University Press, pp. 267-278.

Roth, Alvin E., Tayfun Sonmez, and M. Utku Unver, "Kidney Exchange," *Quarterly Journal of Economics*, 119, 2, May, 2004, 457-488.

Rosen, Harvey S. and Sims, Stephen T. (2011) "Altruistic Behavior and Habit Formation". *Nonprofit Management & Leadership.* 21(3). pp235-253

Royer, Heather, Stehr, Mark and Sydnor, Justin (2013). "Incentives, Commitments and Habit Formation in Exercise: Evidence from a Field Experiment with Workers at a Fortune-500 Company" NBER Working Paper #18580

Schwarz, Norbert. (1998). Accessible content and accessibility experiences: The interplay of declarative and experiential information in judgment. *Personality and Social Psychology Review*, 2, 87-99.

Weber, Elke, Johnson, Eric, Milch, Kerry, Chang, Hannah, Brodscholl, Jeff and Goldstein,Daniel. (2007). "Asymmetric Discounting in Intertemporal Choice: A Query-Theory Account".*Psychological Science* .18,516–23

Weber, E.U. and Johnson, E.J. (2006). Constructing Preferences from Memories. In S.Lichtenstein & P. Slovic (Eds.), New York: Cambridge University Press. pp. 397-410.

Wilson, John (2000) "Volunteering". Annual Review of Sociology. Vol. 26, pp. 215-240

Wilson, J., & Musick, M. (1999). The effects of volunteering on the volunteer. *Law and Contemporary Problems* 62, 141–168

# Appendix

# A. Instructions on the first day.

## Instructions

# (All participants)

Thanks for coming. You have received an information sheet describing the benefits of volunteering. We would like to encourage you to become involved, or to continue to be involved, in volunteering around Pittsburgh.

In particular, we would like to invite you to participate in volunteer activities at the Carnegie Library of Pittsburgh. The next several pages include detailed information about volunteer opportunities at Carnegie Library of Pittsburgh. Please read these pages carefully. Carnegie Library of Pittsburgh is "Free to the People" and volunteers help the library to provide improved services to its user community.

# (Only the Participants in Baseline)

We would also like to invite you share with us your own thoughts on volunteering by emailing us (volunteer-writing@andrew.cmu.edu). In the message, you will write about: 1) why you think volunteering is important for society; and 2) how volunteers can benefit from volunteer activity. Your messages can be based on either your own experience or others' experiences. There is no length requirement for the messages. Only the experimenter will see your messages.

# (Only the Participants in the four incentive treatments)

In order to promote volunteer activities, we will award you with \$25 if you complete all of the tasks below within the next seven days. For example, if today is Tuesday, you need to complete the tasks by next Tuesday.

- 1. Volunteer in at least one of the activities (for at least one hour) listed on the attached page.
- 2. We would also like to invite you to share with us your own thoughts on volunteering by emailing us (volunteer-writing@andrew.cmu.edu) two messages. In the message, you will write about: 1) why you think volunteering is important for society; and 2) how volunteers can benefit from volunteer activity. Your messages can be based on either your own experience or others' experiences. There is no length requirement for the messages, but please note (i) the two messages must be sent to the above email address on different days and (ii) the messages must include different content. Only the experimenter will see your messages.

To pick up your earnings of \$25 please come to this same lab next week at this same time. For example, if today's session is from 2pm-3pm on Tuesday, please come to this lab at 2pm next

Tuesday. If you are not able to come at that particular time, please email the experimenter (<u>experimenter-lab@andrew.cmu.edu</u>) as early as possible to schedule an alternative time.

# (All participants)

To participate in volunteer activities, you should sign up at least one day in advance. <u>Please note</u> that there are limited numbers of volunteer opportunities available each day and it will take up to <u>72 hours for the organizer to respond to your request. Thus</u>, in order to lock-in the day and time that works for you, we recommend that you sign-up as soon as possible.

Please see below the instructions for how to sign-up for volunteering. You are encouraged to fill out the application form and sign-up today so that the volunteer organizer can process all the request forms in a more efficient way. Signing up today also helps you to lock-in the day and time that works for you. You will receive another \$2 if you apply and sign up today.

If you do not apply today, you can still apply and sign-up later. We will email you a link to a Google calendar where you can view the available volunteer activities as well as the sign-up link. You can follow the instructions there to sign-up for volunteering activities any time that is convenient for you.

# Volunteer sign up instructions: (Before signing up to volunteer, please read carefully the attached several pages about the Carnegie Library of Pittsburgh volunteer activities. )

If you are interested in participating, you will need to fill out a short online application form.

For your first volunteer activity at the library, you will be asked to provide your contact information and indicate which program(s), location and time you are available for volunteering. You may choose more than one program, more than one location, more than one time slot, and may indicate your preference over these choices. The volunteer organizer will try to make arrangements based on your preferences. (The application website also provides you instructions about how to contact the library for future sign-up opportunities.)

After receiving your application and sign-up information, the volunteer organizer will email you to confirm the schedule and the location of the activity you signed-up for, and also give you more details regarding how to volunteer. Then you will be on your way to helping our community!

Do you have any question? If so, please raise your hand.

If you have finished reading all the information sheets (volunteer benefit information sheets and the next several pages about the Carnegie Library of Pittsburgh volunteer activities) and instructions, please enter "password" in the box on your computer screen to proceed. You will be asked whether you would like to apply today. If you choose to apply today, please choose "Yes"

and fill out the information. If you choose not to apply today, please choose "No". Again, by applying today you receive a \$2 bonus for today's experimental earnings.

# B. Information about volunteering opportunities at CLP

# Carnegie Library of Pittsburgh Volunteer Opportunities

Thank you for your interest in volunteering for Carnegie Library of Pittsburgh. Volunteers provide an invaluable service to the librarians and the public. By shelving and caring for library materials, or sorting requested items destined for each of the 74 public libraries in Allegheny County, you free librarians to do collection development, reference work, program presentation and outreach. Without our volunteers we would have less time to work with the children, teens and adults, and our materials would not be as accessible for our customers.

For any questions, please email:

Karen Meharra or Kaylee Petsch Clpvol+volunteer@gmail.com Volunteer Services Carnegie Library of Pittsburgh

Below is a list of volunteer opportunities the library needs assistance with:

# **Adopt-A-Shelf Volunteer Program**

# **Locations and Times**

- Main Library First Floor Department- Monday-Thursday 10:30-7:30; Friday-Saturday 10:30- 5:00; Sunday 12:30-4:30
- Main Library Music Department-Monday-Thursday 10:30-7:30; Friday-Saturday 10:30- 5:00; Sunday 12:30-4:30
- East Liberty- Monday-Wednesday 10:30-7:30; Thursday-Saturday 10:30-4:30
- Homewood- Monday-Tuesday and Friday-Saturday 10:30-4:30; Wednesday-Thursday 10:30-6:30

Customers and staff rely upon volunteers to have the books in the proper place and in good condition so that library materials can be used. The aim of the Adopt-a Shelf Program is to create a convenient way for volunteers to be able to help better maintain the stacks at the Carnegie Library of Pittsburgh.

Volunteers will be assigned a section of library stacks and shelf read to make sure the books in your adopted section are in proper order, arranged neatly, properly stickered, clean, and undamaged. Volunteers of any age, or even families can apply for this activity.

The Adopt-a-Shelf Program has the potential of ensuring that more sections of the library are properly maintained, while also making volunteers feel that they are an important part of the Library because part of the library is "theirs".

# Directions

Main Library- 4400 Forbes Ave. Pittsburgh, PA. Can take the 61A, 61B, 61C, 61D public buses to Forbes and Craig St. Library is right next to the Natural History museum.

East Liberty Library- 130 S. Whitfield St. Pittsburgh, PA. Can take the75 Outbound public bus to Spahr St. Then walk about a 5 minute walk to the library.

Homewood Library – 7101 Hamilton Drive Pittsburgh, PA. Can Take the 71D outbound public bus to Hamilton and Lang. Then the library is next to stop.



# **Location and Times**

- Delivery Services, also known as Shipping, is located at the Library Support Center, 50 Alexander Street, Pittsburgh, PA 15220 (West End).
  - There is a free parking lot, adjacent to the Library Support Center.
- Volunteers are welcome to help at Shipping on Monday Friday 9:00am to 4:00pm any amount of time is appreciated

Please help Allegheny County's Public Library Support Center sort library materials in preparation for delivery to the over 74 public libraries in Allegheny County. Last year over 4 million library items went through Shipping. Without volunteers like you we would not be able to meet the needs of Allegheny County residents.

Five times per week van delivery service is provided for all public libraries within Allegheny County. This service provides the transfer of books, audiovisual materials and mail.

Volunteers work with friendly library staff in a team environment sorting books and other library materials to prepare them for delivery. The Shipping area is well lit, comfortable (albeit on a concrete floor), with a Crystal Springs Water Cooler. The task requires standing, bending, and lifting books that can weigh up to 5 lbs.

# Directions

Library Support Center: 50 Alexander Street, Pittsburgh, PA. Take the 61D inbound bus to 6<sup>th</sup> Ave and Wood St. Then walk to 7<sup>th</sup> Ave and Penn Ave and take the 31 outbound bus to S. Main St and Wabash. Then about a 2 minute walk to the Library Support Center.

# **Street Teams Marketing Library Programs**

# **Locations and Times**

- Various neighborhoods in Pittsburgh—(Squirrel Hill, Oakland, East Liberty, Regent Square/Point Breeze, South Side, Strip District, Shadyside, Bloomfield)
- Anytime at volunteers convenience

Explore neighborhoods and talk with the businesses and organizations owners/managers regarding hanging posters and distributing flyers advertising programs at the Carnegie Library of Pittsburgh - Main (Oakland).

Carnegie Library of Pittsburgh is "Free to the People." We would appreciate help in spreading the word about resources and programs at Carnegie Library of Pittsburgh.

# C. Messages to the participants in the four incentive treatments after the \$25 week

(*All participants in the VW1, V, VW and W treatments*) Please read the following information carefully.

We would like to encourage you to continue to be involved in volunteering around Pittsburgh. In particular, we would like to invite you to continue participating in volunteer activities at the Carnegie Library of Pittsburgh. Carnegie Library of Pittsburgh is "Free to the People" and volunteers help the library to provide improved services to its user community.

Please continue to use the Google calendar to sign up for volunteer activities.

# (Participants assigned to V treatment)

In order to continue promoting volunteer activities, we will award you with \$50 if you complete the tasks below within the next 28 days.

Volunteer for the activities listed on the Google calendar at least four times (for at least one hour each time) within the following 28 days. For example, if today is Tuesday (Sept. 18), you need to volunteer at least four times by the Tuesday of the fourth week (Oct. 16). Although you can decide which days to volunteer, <u>we encourage you to</u> <u>volunteer at least once in each of the following four seven-day cycles.</u> For example, if today is Tuesday, we encourage you to volunteer at least once by Tuesday of each of the following four weeks.

# (Participants assigned to VW treatment)

In order to continue promoting volunteer activities, we will award you with \$50 if you complete the tasks below within the next 28 days.

1. Volunteer for the activities listed on the Google calendar at least four times (for at least one hour each time) within the following 28 days. For example, if today is Tuesday (Sept. 18), you need to volunteer at least four times by the Tuesday of the fourth week (Oct. 16). Although you can decide which days to volunteer, we encourage you to volunteer at least once in each of the following four seven-day cycles. For example, if today is Tuesday, we encourage you to volunteer at least once by Tuesday of each of the following four weeks.

2. Email us (volunteer-writing@andrew.cmu.edu) two messages during each of the following four seven-day cycles. For example, if today is Tuesday, you need to email two messages <u>by Tuesday of each of the following four weeks</u>. In each message, you will write about: 1) why you think volunteering is important for society; and 2) how volunteers can benefit from volunteer activity. Your messages can be based on either your own experience or others' experiences. There is no length requirement for the messages, but please note (i) the two messages must be sent to the above email address on different days and (ii) the messages must include different content. Only the experimenter will see your messages.

# (Participants assigned to W treatment)

In order to continue promoting volunteer activities, we will award you with \$50 if you complete the tasks below within the next 28 days.

Email us (volunteer-writing@andrew.cmu.edu) three messages during each of the following four seven-day cycles. For example, if today is Tuesday, you need to email three messages <u>by Tuesday of each of the following four weeks</u>. In each message, you will write about: 1) why you think volunteering is important for society; and 2) how volunteers can benefit from volunteer activity. Your messages can be based on either your own experience or others' experiences. There is no length requirement for the messages, but please note (i) the three messages must be sent to the above email address on different days and (ii) the messages must include different content. Only the experimenter will see your messages.

# (All participants in the V, VW and W treatments)

To pick up your earnings of \$50 please come to this same lab at this same day and time four weeks from now. For example, if today's session is between 2pm and 3pm on Sept. 18 (Tuesday), please come to this lab between 2pm and 3pm on Oct. 16 (Tuesday). If you are not able to come at that particular time, please email the experimenter as early as possible to schedule an alternative time on that day.

# (All participants in the VW1, V, VW and W treatments)

Again, to participate in the volunteer activities, you should sign up at least 24 hours in advance. In your sign up email (clpvol+volunteer@gmail.com), **please provide specific date(s), time and program(s) for which you would like to volunteer**.

<u>Please note that there are limited numbers of volunteer opportunities available each day and it</u> <u>will take up to 72 hours for the organizer to respond to your request. Thus,</u> in order to lock-in the day and time that works for you, we recommend that you sign up as soon as possible.

Note: We will email you an electronic copy of this information sheet.